Author's response to reviews

Title: Community-based mental health treatments for survivors of torture and militant attacks in Southern Iraq: a randomized control trial

Authors:

William M Weiss (wweiss1@jhu.edu)
Laura K Murray (lmurra15@jhu.edu)
Goran A Zangana (gsabir@heartlandalliance.org)
Zayan Mahmooth (zayanmahmooth@gmail.com)
Debra Kaysen (dkaysen@u.washington.edu)
Shannon Dorsey (dorsey2@u.washington.edu)
Kristen Lindgren (kpl9716@uw.edu)
Alden Gross (agross14@jhu.edu)
Sarah M Murray (smurray9@jhu.edu)
Judith K Bass (jbass1@jhu.edu)
Paul Bolton (pbolton1@jhu.edu)

Version: 5
Date: 23 September 2015

Author's response to reviews:

We have included figure legends for figure 1 after the References.

We removed the tracked changes copy of the manuscript from the additional files.

We removed the study protocol from the additional files.